

G OPEN ACCESS

Citation: Zagkos L, Dib M-J, Pinto R, Gill D, Koskeridis F, Drenos F, et al. (2022) Associations of genetically predicted fatty acid levels across the phenome: A mendelian randomisation study. PLoS Med 19(12): e1004141. https://doi.org/10.1371/ journal.pmed.1004141

Academic Editor: Ronald C. W. Ma, Chinese University of Hong Kong, CHINA

Received: June 16, 2022

Accepted: November 18, 2022

Published: December 29, 2022

Copyright: © 2022 Zagkos et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: UK Biobank individual level data used in this work can be accessed after applying for access at (https://www. ukbiobank.ac.uk/enable-your-research/apply-foraccess). Summary statistics used in the analysis can be openly accessed at the IEU OpenGWAS project (https://gwas.mrcieu.ac.uk/). Additional links to datasets used in our work: UKB fatty acids: https://gwas.mrcieu.ac.uk/ FinnGen dataset: https://finngen.gitbook.io/documentation/v/r7/ GIANT consortium: http://ftp.ebi.ac.uk/pub/ databases/gwas/summary_statistics/ RESEARCH ARTICLE

Associations of genetically predicted fatty acid levels across the phenome: A mendelian randomisation study

Loukas Zagkos^{1*}, Marie-Joe Dib¹, Rui Pinto^{1,2}, Dipender Gill^{1,3,4}, Fotios Koskeridis⁵, Fotios Drenos^{6,7}, Georgios Markozannes⁵, Paul Elliott^{1,2,8}, Verena Zuber^{1,2}, Kostas Tsilidis^{1,5}, Abbas Dehghan^{1,2}, Ioanna Tzoulaki^{1,2,5,8}

Department of Epidemiology and Biostatistics, School of Public Health, Imperial College London, London, United Kingdom, 2 UK Dementia Research Institute, Imperial College London, London, United Kingdom,
Chief Scientific Advisor Office, Research and Early Development, Novo Nordisk, Copenhagen, Denmark,
Medical Research Council Biostatistics Unit, University of Cambridge, Cambridge, United Kingdom,
Department of Hygiene and Epidemiology, University of Ioannina Medical School, Ioannina, Greece,
Department of Life Sciences, College of Health, Medicine and Life Sciences, Brunel University London,
Uxbridge, United Kingdom, 7 Institute of Cardiovascular Sciences, University College London, London,
United Kingdom, 8 BHF Centre of Excellence at Imperial College London, London, United Kingdom

* l.zagkos@imperial.ac.uk

Abstract

Background

Fatty acids are important dietary factors that have been extensively studied for their implication in health and disease. Evidence from epidemiological studies and randomised controlled trials on their role in cardiovascular, inflammatory, and other diseases remains inconsistent. The objective of this study was to assess whether genetically predicted fatty acid concentrations affect the risk of disease across a wide variety of clinical health outcomes.

Methods and findings

The UK Biobank (UKB) is a large study involving over 500,000 participants aged 40 to 69 years at recruitment from 2006 to 2010. We used summary-level data for 117,143 UKB samples (base dataset), to extract genetic associations of fatty acids, and individual-level data for 322,232 UKB participants (target dataset) to conduct our discovery analysis. We studied potentially causal relationships of circulating fatty acids with 845 clinical diagnoses, using mendelian randomisation (MR) approach, within a phenome-wide association study (PheWAS) framework. Regression models in PheWAS were adjusted for sex, age, and the first 10 genetic principal components. External summary statistics were used for replication. When several fatty acids were associated with a health outcome, multivariable MR and MR-Bayesian method averaging (MR-BMA) was applied to disentangle their causal role. Genetic predisposition to higher docosahexaenoic acid (DHA) was associated with cholelithiasis and cholecystitis (odds ratio per mmol/L: 0.76, 95% confidence interval: 0.66 to 0.87). This was supported in replication analysis (FinnGen study) and by the genetically

GCST006001-GCST007000/GCST006900/ IGAP consortium: http://ftp.ebi.ac.uk/pub/databases/ gwas/summary_statistics/GCST002001-GCST003000/GCST002245/

CARDIoGRAMplusC4D cohort: http://ftp.ebi.ac.uk/ pub/databases/gwas/summary_statistics/ GCST003001-GCST004000/GCST003116/ MAGIC cohort: http://ftp.ebi.ac.uk/pub/databases/gwas/ summary_statistics/GCST000001-GCST001000/ GCST000803/ DIAGRAM cohort: https://diagramconsortium.org/downloads.html Kettunen et al., 2016 dataset: https://www.ebi.ac.uk/gwas/ publications/27005778.

Funding: This work was supported by Health Data Research UK, which is funded by the UK Medical Research Council. Engineering and Physical Sciences Research Council, Economic and Social Research Council, Department of Health and Social Care (England), Chief Scientist Office of the Scottish Government Health and Social Care Directorates, Health and Social Care Research and Development Division (Welsh Government), Public Health Agency (Northern Ireland), British Heart Foundation and Wellcome (Grant Number CFC1007). PE is the director of the Medical Research Council (MRC) Centre for Environment and Health and acknowledges support from the MRC (MR/L01341X/1: MR/ S019669/1). PE and IT also acknowledge support from the Imperial College British Heart Foundation Centre for Research Excellence (RE/18/4/34215), and the UK Dementia Research Institute at Imperial College London (MC PC 17114), and Health Data Research UK for London. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: DG is employed part-time by Novo Nordisk. VZ is a paid statistical consultant on PLOS Medicine's statistical board.

Abbreviations: BMI, body mass index; CHD, coronary heart disease; CVD, cardiovascular disease; DHA, docosahexaenoic acid; EPA, eicosapentaenoic acid; FA, fatty acid; FDR, false discovery rate; GRS, genetic risk score; GWAS, genome-wide association study; HES, Hospital Episode Statistics; HWE, Hardy–Weinberg equilibrium; IVW, inverse-variance weighted; LA, linoleic acid; LD, linkage disequilibrium; LDSC, linkage disequilibrium score regression; MAF, minor allele frequency; MR, mendelian randomization; MR-BMA, MR, Bayesian method averaging; MUFA, monounsaturated fatty acid; NMR, nuclear magnetic resonance; OR, odds ratio; predicted omega-3 fatty acids analyses. Genetically predicted linoleic acid (LA), omega-6, polyunsaturated fatty acids (PUFAs), and total fatty acids (total FAs) showed positive associations with cardiovascular outcomes with support from replication analysis. Finally, higher genetically predicted levels of DHA (0.83, 0.73 to 0.95) and omega-3 (0.83, 0.75 to 0.92) were found to have a protective effect on obesity, which was supported using body mass index (BMI) in the GIANT consortium as replication analysis. Multivariable MR analysis suggested a direct detrimental effect of LA (1.64, 1.07 to 2.50) and omega-6 fatty acids (1.81, 1.06 to 3.09) on coronary heart disease (CHD). MR-BMA prioritised LA and omega-6 fatty acids as the top risk factors for CHD. Although we present a range of sensitivity analyses to the address MR assumptions, horizontal pleiotropy may still bias the reported associations and further evaluation in clinical trials is needed.

Conclusions

Our study suggests potentially protective effects of circulating DHA and omega-3 concentrations on cholelithiasis and cholecystitis and on obesity, highlighting the need to further assess them as prevention treatments in clinical trials. Moreover, our findings do not support the supplementation of unsaturated fatty acids for cardiovascular disease prevention.

Author summary

Why was this study done?

- Epidemiological studies have investigated the influence of dietary fatty acids on several chronic diseases including cardiovascular disease (CVD), inflammatory diseases, cancers, and neurodegenerative conditions.
- The evidence is conflicting and inconclusive and is not supported by randomised controlled trials (RCTs) on fatty acids supplementation.
- The fact that the human fatty acid metabolome partly reflects the fatty acid intake was leveraged in this work.

What did the researchers do and find?

- Mendelian randomisation analysis was conducted to investigate potential causal associations of circulating fatty acids and 845 disease outcomes in UK Biobank.
- We found that 1 mmol/L increase in genetically predicted docosahexaenoic acid (DHA) was associated with 24% reduced odds of cholelithiasis and cholecystitis, as well as 17% reduced odds of obesity.
- One mmol/L higher genetically predicted linoleic acid (LA) was associated with 27% increased odds of ischemic heart disease.

PheWAS, phenome-wide association study; PUFA, polyunsaturated fatty acid; RCT, randomised controlled trial; SFA, saturated fatty acid; SNP, single nucleotide polymorphism; T2D, type 2 diabetes; UKB, UK Biobank.

• When assessed with other fatty acid classes simultaneously, 1 mmol/L higher genetically predicted LA and omega-6 fatty acids were associated with higher odds of coronary heart disease (64% and 81% increase, respectively).

What do these findings mean?

- Our results support the prioritisation of RCTs examining the role of primary and secondary prevention of cholelithiasis via dietary modification or supplementation of DHA and other omega-3 fatty acids.
- Our principal findings also provide evidence against the supplemental use of fatty acids for CVD prevention.
- Triangulation of evidence for causal inference of fatty acid levels through experimental studies susceptible to different assumptions is needed to strengthen the evidence.

Introduction

Fatty acids (FAs) are major constituents of several lipid species and are involved in diverse metabolic pathways and biochemical processes in normal cells [1]. Their role in health and disease has been extensively studied; however, the mechanisms by which fatty acids may exert their effects on health outcomes are still not fully understood, and the causality between different fatty acids and health outcomes has not been established. Numerous epidemiological studies have investigated the quantity and quality of dietary fatty acids in relation to several chronic diseases including cardiovascular disease (CVD), cancers, neurodegenerative conditions, and inflammatory diseases [2,3]. However, this evidence remains inconclusive [4], and it is often not supported by randomised controlled trials (RCTs) on fatty acid supplementation, which have also shown contradictory results [5–8]. There is, therefore, an urgent need to establish the potential causal role of fatty acids in relation to a wide range of phenotypes as fatty acid supplements are the most commonly consumed nonvitamin/nonmineral dietary supplements by both adults and children in western societies [9,10].

The human fatty acid metabolome partly reflects the dietary fatty acid intake and has the potential to explain their pathophysiological roles. Circulating fatty acid levels are influenced by environmental and genetic factors [11–15] and can be used as biomarkers of fatty acid intake. The present study aimed to assess potentially causal associations between circulating fatty acid levels and the risk of different health outcomes using mendelian randomisation (MR) analysis. MR analysis uses genetic polymorphisms that are randomised by nature and can overcome the limitations of traditional epidemiological studies, as the random allocation of alleles during conception minimises confounding, and their presence from birth prevents reverse causation. Our study leveraged recent genetic data generated from over 115,000 UK Biobank (UKB) participants on circulating levels of 8 nuclear magnetic resonance (NMR) fatty acid measures, providing unprecedented power for fatty acids MR studies. Specifically, we investigated associations for docosahexaenoic acid (DHA) and omega-3 fatty acids, linoleic acid (LA), and omega-6 fatty acids, as well as the fatty acid classes: monounsaturated fatty acids (MUFAs), polyunsaturated fatty acids (PUFAs), saturated fatty acids (SFAs), and total

fatty acids (total FAs). We implemented a systematic approach to examine causal pathways between each fatty acid trait and the human phenome, followed by multivariable MR and the MR-Bayesian model averaging (MR-BMA) framework to identify direct effects of different fatty acids and to select the ones that are the most likely causal risk factors for disease phenotypes.

Methods

Study design

The study design is shown in **Fig 1**. This study used a phenome-wide association study (Phe-WAS) to identify associations between genetically predicted fatty acid concentrations and several diseases in UKB. For statistically significant associations, a two-sample MR of summarylevel data was conducted to further investigate the causal role of fatty acids on health outcomes followed by multivariable MR and MR-BMA. We further attempted to replicate statistically significant MR associations using alternative data sources for the exposures (fatty acids) and the outcomes. This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline, specific for MR (S1 Checklist).

Data sources

UK Biobank. The UKB [16] is a large ongoing prospective study involving over 500,000 participants aged 40 to 69 years at recruitment from 2006 to 2010. The study has collected



Fig 1. Overview of the study. BMI, body mass index; CHD, coronary heart disease; CVD, cardiovascular disease; DHA, docosahexaenoic acid; FDR, false discovery rate; GRS, genetic risk score; GWAS, genome-wide association study; HbA1c, haemoglobin A1c; HES, Hospital Episode Statistics; HWE p, Hardy–Weinberg equilibrium *p*-value; IVW, inverse-variance weighted; LA, linoleic acid; MAF, minor allele frequency; MR-PheWAS, mendelian randomisation–phenome-wide association study; MUFA, monounsaturated fatty acid; NMR, nuclear magnetic resonance; *P, p*-value; PUFA, polyunsaturated fatty acid; SFA, saturated fatty acid; total FA, total fatty acid; T2D, type 2 diabetes.

https://doi.org/10.1371/journal.pmed.1004141.g001

biological samples and a wide range of phenotypic data from its participants, including data from questionnaires, physical measures, sample assays, genome-wide genotyping, and longitudinal follow-up for a plethora of health-related outcomes. Details of genotyping and data management have been described [17].

Recently, UKB released high-throughput NMR metabolic biomarkers for a random subset of 117,994 participants at baseline assessment including measurements of 8 circulating fatty acids: DHA, LA, MUFAs, PUFAs, omega-3, omega-6, SFAs, and total FAs.

Fatty acids genetic instruments. We obtained genetic summary-level data from genomewide association studies (GWAS) for the 8 measured fatty acids on 114,999 European UKB participants from the MRC IEU OpenGWAS project [18]. Single nucleotide polymorphisms (SNPs) with a minor allele frequency (MAF) greater than 0.1 and a Hardy–Weinberg equilibrium (HWE) *P* value greater than 10^{-6} were considered. Multiallelic SNPs were excluded from the analysis. Independent SNPs associated with each of the 8 fatty acid concentrations were identified at a genome-wide significance threshold ($P < 5 \times 10^{-8}$). Independence of SNPs was assessed through clumping for linkage disequilibrium (LD) $r^2 < 0.001$ using a window of 10 million bases and the 1,000 genomes phase 3 European reference panel through the opensource association toolset plink [19]. In the replication phase, we used the Kettunen and colleagues' GWAS [20] as an alternative source of beta estimates for genetic variants of the same fatty acids using the same criteria as above. To rule out exogenous factors that may affect the genetic associations with the fatty acids, we excluded 13,577 omega-3 supplementation and/or lipid lowering drug users with available metabolic profiles in UKB, recalculated the fatty acid genetic associations, and monitored the percentage change in genetic association estimates.

Health outcomes. For the PheWAS analysis, individual-level data from UKB were used for clinical health outcomes with more than 200 cases, obtained from national registries, such as Hospital Episode Statistics (HES), cancer registry, and death registry. Cases were defined according to ICD-9 and ICD-10 codes (International Classification of Diseases, ninth and 10th revisions). We also used alternative (independent) GWAS sources for the outcomes, which showed statistically significant associations in the two-sample MR analyses. We selected the largest published GWAS for each outcome, and when there was no available GWAS, we used FinnGen (https://r5.finngen.fi/) estimates. In detail, we used GIANT consortium for body mass index (BMI) [21], CARDIoGRAMplusC4D for coronary heart disease (CHD) [22], IGAP [23] for Alzheimer's disease, MAGIC [24] for HbA1c and DIAGRAM [25] for type 2 diabetes (T2D), and FinnGen (https://r5.finngen.fi/) for cholelithiasis and "other biliary tract disease".

Statistical analysis

Genetic risk score generation. A weighted genetic risk score (GRS) was generated for each fatty acid concentration for UKB participants by adding up the number of risk-increasing alleles for all the variants selected as instruments. The number of risk-increasing alleles at each SNP was multiplied by the effect estimate of the association of that SNP with the exposure. This analysis was performed in the "target" UKB dataset, which excludes individuals with available NMR metabolic biomarker data (base dataset), so as to avoid sample overlap between the base dataset (the 117,143 UKB samples used for the estimation of the effect of SNPs on the exposures) and the target dataset (the 322,232 UKB samples used for the generation of the GRS, after excluding the base dataset). For the GRS generation in the target dataset, we excluded non-European participants, one randomly selected participant from each pair of up to third-degree relatives (kinship coefficient > 0.0884) [17] and individuals with discordant reported sex and genetic sex.

Phenome-wide association study. A PheWAS aims to investigate possible associations between a single GRS or genotype and multiple phenotypes. We conducted 8 PheWAS to investigate potential associations between the GRS of each specific fatty acid measurement and 845 clinical diagnoses. All regression models were adjusted for sex, age, and the first 10 genetic principal components, estimated by UKB (Data-Field 22009). To account for type 1 error inflation due to multiple testing, we used the false discovery rate (FDR) method, controlled at 5% significance level by the Benjamini–Hochberg procedure [26].

Two-sample mendelian randomisation and sensitivity analyses. For the associations reaching 5% FDR in the GRS-PheWAS analysis, we performed two-sample MR analysis. MR assesses causal relationships using genetic variants as instrumental variables. A genetic variant can be considered as an instrumental variable if it satisfies three assumptions: it is robustly associated with the exposure, it is independent of any confounders of the exposure-outcome relationship, and it is associated with the outcome only via the exposure. To obtain the effect estimates of the associations between the genetic instruments and the health outcomes, we fitted logistic regression models in UKB, adjusting for sex, age, and the first 10 genetic principal component. In this MR framework, there was no sample overlap between the UKB populations used for the fatty acids (base dataset) and for the clinical diagnoses (target dataset). The main analysis was conducted using the random-effects inverse-variance weighted (IVW) method, which provides precise causal estimates, assuming that all variants are valid instrumental variables [27]. In sensitivity analyses, we applied the MR-Egger method to explore the assumption of no pleiotropy in genetic instruments [28]. In addition, we performed the MRweighted median method, which returns an accurate causal estimate, provided that at least 50% of the weight in the analysis comes from valid instrumental variables [29]. The I² statistic was calculated to detect heterogeneity among the MR estimates obtained from multiple genetic variants. Last, we used MR-PRESSO, which detects and removes pleiotropic variants based on their contributions to heterogeneity [30]. We considered associations to be statistically significant if P values in MR-IVW, MR-weighted median, and MR-PRESSO methods were smaller than 0.05, with supportive MR-Egger. MR-Egger was considered supportive when the effect estimate was in the same direction as MR-IVW and the MR-Egger intercept was statistically nonsignificant (P > 0.05). The F-statistic for each genetic instrument was generated to assess their strength. We included instruments with F-statistic > 10 in the analysis. Moreover, as a sensitivity analysis for the fatty acid associations with CHD, BMI, and cholelithiasis, we excluded the genetic instruments within 500 kb of the FADS locus (chromosome 11: 61,067,097 to 62,134,826).

To assess the robustness of our findings, we attempted to replicate findings using alternative data sources for the exposure and the outcome, respectively. We performed the same sequence of MR analyses (main and sensitivity analyses described above) using Kettunen and colleagues' GWAS [20] for fatty acids and the largest available published GWAS for each outcome of interest (see Health outcomes above).

All estimates were reported as odds ratio (OR) per 1 mmol/L increase in fatty acid concentrations, together with their 95% confidence interval.

Multivariable mendelian randomisation. To disentangle the relationships between correlated circulating fatty acids and CHD, we conducted two analytical approaches. Firstly, we conducted multivariable MR, a statistical method for estimation of causal effects using genetic variants associated with more than one risk factor [31]. Multivariable MR allows an estimation of the direct effect of each fatty acid on CHD, that is, an effect that is not mediated by any other fatty acid in the model. Secondly, we conducted MR-BMA, an extension of multivariable MR, which selects and prioritises candidate highly correlated risk factors with shared genetic predictors using a Bayesian framework as previously described [32]. For MR-BMA, we

considered each set of fatty acids in turn, and for each set of fatty acids, we undertook a multivariable MR analysis using weighted regression based on summarised genetic data. Regression models were assessed using goodness of fit, and a score was assigned to each set of fatty acids that represented the model's posterior probability of that set being the true causal risk factors. Then, for each of the candidate fatty acids, we summed up the posterior probability over models including that fatty acid to compute its marginal inclusion probability, representing its probability of being a causal determinant of disease risk. We also calculated the model-averaged causal effect, which represents the average causal effect across models including that fatty acid. For all instruments, we generated the conditional F-statistic to assess their strength [33]. Using both approaches, we fitted five models based on the hierarchical order of the fatty acid measurements, considering different combinations to avoid overlap between fatty acids (e.g., DHA and omega-3). We fitted a model with DHA, LA, MUFA, and SFA; a second model considering omega-3, omega-6, MUFA, and SFA; and a third with PUFA, MUFA, and SFA. Additionally, we assessed the independent effects of PUFAs alone on CHD, i.e., DHA and LA (model 4), and omega-3 with omega-6 (model 5).

Statistical software. Analysis was conducted in R version 4.0.2 [34], phenome-wide associations were conducted using R package "PheWAS" [35], two-sample MR was performed using "TwoSampleMR" [36] and "MRPRESSO" [30] R packages, and MR-BMA was conducted using the methodology provided in [32]. Figures were produced using the R package "forestplot" [37].

Results

Fatty acid GRS was calculated on UKB participants, after excluding those with available NMR metabolic biomarker data, to avoid sample overlap between the base dataset (the 117,143 UKB samples used for the estimation of the effect of SNPs on the exposures) and the target dataset (the 322,232 UKB samples used for the generation of the GRS, after excluding the base dataset).

Descriptive characteristics of the 117,143 UKB samples (base dataset) and 322,232 UKB samples (target dataset) used in the discovery phase and the cohorts used in the replication phase are provided in S1 Table. Aside from the phenotypic correlations between DHA and LA (Pearson's R = 0.09), the remaining correlations between fatty acid measurements were relatively high (highest correlation 0.97 between total FA and SFA). Genetic correlations between the fatty acid measurements were estimated using linkage disequilibrium score regression (LDSC) [38] (S2 Table) and showed a similar pattern as the phenotypes. As expected, the genetic instruments for each fatty acid shared a number of SNPs, especially those belonging in nested groups, e.g., DHA and omega-3. The hierarchical order of the fatty acid measurements, phenotypic and genetic correlations between the fatty acids and the genetic variants used as instruments in the analysis are provided in Fig 2. The list of independent genetic variants used as instrumental variables in this analysis and their corresponding F-statistic are provided in S3 and S4 Tables.

Phenome-wide association study

PheWAS examined 845 distinct clinical diagnoses grouped into 17 disease categories (median number of cases: 879 [range: 200 to 47,166]) (S5 Table). Thirty-five unique clinical outcomes were associated with at least one fatty acid GRS at 5% FDR threshold ($P = 9.2 \times 10^{-4}$); 9 of these outcomes were within the CVD group (positive associations) and 5 within the digestive group (negative associations) (Fig 3; see full results in S6–S13 Tables and S1–S8 Figs).



Fig 2. (1a) Pearson correlation coefficient heatmap between phenotypic circulating fatty acid measurements. (1b) Genetic correlation heatmap between the fatty acid measurements. (2) Additional information on the genetic instruments used in the analysis. Independent biallelic genetic variants that are associated with fatty acids below $P = 5 \times 10^{-8}$, have MAF > 0.1 and HWE $P > 10^{-6}$ were used as instruments. Independence was assessed through clumping with $r^2 < 0.001$ and 10 million bases clumping window. Dark red horizontal bars at the left bottom corner show the total number of genetic variants used as instruments per fatty acid. Dark blue vertical bars depict the number of genetic variants within each intersection. SNP intersections are defined with the purple dots below the vertical bars, indicating SNP overlap between groups, e.g., the first bar refers to MUFA having 23 unique SNPs, the sixth bar refers to 10 common SNPs between DHA and omega-3 etc. (3) Hierarchical order of the fatty acids. DHA, docosahexaenoic acid; HWE, Hardy–Weinberg equilibrium; LA, linoleic acid; PUFA, polyunsaturated fatty acid; SFA, saturated fatty acid; SNP, single nucleotide polymorphism; total FA, total fatty acid.

https://doi.org/10.1371/journal.pmed.1004141.g002

Mendelian randomisation

MR results are summarised in <u>S14 Table</u>. Out of 114 MR tests, 60 associations were statistically significant, following the criteria described in the Methods section. Phenotypes with supporting evidence of causality are discussed below.

Higher genetically predicted levels of DHA were found to have a protective effect on cholelithiasis and cholecystitis (OR per 1 mmol/L increase in DHA: 0.76, 95% CI: 0.66 to 0.87) and cholelithiasis (0.77, 0.67 to 0.89) (Fig 4). These were supported in replication analysis (Finn-Gen study as the outcome GWAS) and by the omega-3 FA analyses and were consistent after SNPs in the pleiotropic FADS region were excluded (<u>\$15</u> and <u>\$16</u> Tables).

Genetically predicted LA, omega-6, PUFA, and total FA showed positive associations with cardiovascular outcomes. More specifically, we identified an increased risk on ischemic heart disease for higher total FA (1.27, 1.05 to 1.54), PUFA (1.25, 1.03 to 1.51), omega-6 (1.28, 1.04 to 1.58), and LA (1.27, 1.03 to 1.56) per 1 mmol/L increase in genetically predicted levels, as seen in **Fig 5**. These associations were consistent in replication analysis using CARDIoGRAM-plusC4D as the outcome GWAS source (**S17 Table**). After excluding SNPs in the FADS locus, all fatty acids retained their significant detrimental effects on CHD, except for DHA and



Fig 3. PheWAS results for the associations between 8 weighted GRS of different fatty acids and 845 clinical outcomes in UKB. log(OR) regression estimates are plotted against -log(P value). The dashed line represents the 5% FDR threshold, which corresponds to $P = 9.2 \times 10^{-4}$. Due to their high *P* values, associations with disorders of lipid metabolism, hypercholesterolemia, and hyperlipidaemia can be seen in the top left box. DHA, docosahexaenoic acid; FDR, false discovery rate; GRS, genetic risk score; LA, linoleic acid; MUFA, monounsaturated fatty acid; PheWAS, phenome-wide association study; PUFA, polyunsaturated fatty acid; SFA, saturated fatty acid; total FA, total fatty acid; UKB, UK Biobank.

https://doi.org/10.1371/journal.pmed.1004141.g003

omega-3 (S16 Table). Considerable heterogeneity was observed in these analyses (I² statistic > 50%), which, when reduced by removing outliers from MR-PRESSO, resulted in effect estimates supporting a causal effect. Replication analysis using Kettunen and colleagues' betas for the fatty acids further supported the positive association for genetically predicted LA, omega-6 fatty acids, and PUFA in relation to CHD (S15 Table).

Higher genetically predicted levels of DHA and omega-3 were found to have a protective effect on obesity (0.83, 0.73 to 0.95) and (0.83, 0.75 to 0.92), respectively (**Fig 6**). This was further supported in replication analysis using BMI as the outcome of interest (GIANT consortium) (**S15 Table**). These associations with BMI were attenuated to the null when excluding SNPs in the FADS locus (**S16 Table**). Moreover, higher levels of genetically predicted total FA concentration had a protective effect on osteoarthrosis (1.27, 1.02 to 1.57), which was not supported in replication analysis using osteoarthritis as the outcome of interest in FinnGen.

Multivariable mendelian randomisation

Five multivariable models were fitted to examine the direct effects of fatty acids on CHD, according to their hierarchical order (S18 Table). Increasing LA retained a potentially causal

Cholelithiasis, cholecystitis and 'other biliary tract disease'

📕 MR-IVW 📕 MR-WM 📒 MR-Egger 📕 MR-PRESSO



Fig 4. Two-sample MR effect estimates per 1 mmol/L higher fatty acids for cholelithiasis, cholecystitis, and "other biliary tract disease" in (a) using betas from UKB and (b) using betas from Kettunen and colleagues [20] for fatty acids and FinnGen (https://r5.finngen.fi/) for cholelithiasis. Only associations between genetically predicted fatty acids and biliary tract health outcomes with statistically significant (P < 0.05) IVW MR estimates in UKB are shown. DHA, docosahexaenoic acid; IVW, inverse-variance weighted; MR, mendelian randomisation; UKB, UK Biobank; WM, weighted-median.

https://doi.org/10.1371/journal.pmed.1004141.g004

relationship with CHD when assessed together with DHA, MUFA, and SFA (1.64, 1.07 to 2.50) and when assessed with DHA only (1.32, 1.07 to 1.64). Higher omega-6 was also associated with increased risk of CHD, when omega-3, omega-6, MUFA, and SFA were simultaneously fitted into a model (1.81, 1.06 to 3.09) and when fitted into a model with omega-3 only (1.33, 1.01 to 1.75) (Fig 7). In the MR-BMA analysis, LA was narrowly the top-ranked risk factor for CHD in the model with DHA, MUFA, and SFA (marginal inclusion probability, 0.31) and when assessed with DHA (marginal inclusion probability, 0.55), while omega-6 was identified as the prioritised risk factor for CHD (marginal inclusion probability, 0.31) when assessed with omega-3, MUFA, and SFA and (marginal inclusion probability, 0.55) in the model with omega-6.

Discussion

In this study, we used MR to explore and compare the relationships of fatty acids levels with a wide range of clinical phenotypes across the phenome. Our principal findings include the positive associations between several genetically predicted fatty acids and CVD outcomes with multivariable analysis highlighting omega-6 fatty acids and LA in particular, having an independent causal positive effect on CHD. This adds further evidence against the supplemental

Fatty acid	Outcome	N cases	
DHA (UKB)	IHD (UKB)	22242	
DHA (Kettunen et al.)	CHD (CARDIoGRAMplusC4D)	63746	
LA (UKB)	IHD (UKB)	22242	
LA (Kettunen et al.)	CHD (CARDIoGRAMplusC4D)	63746	
MUFA (UKB)	IHD (UKB)	22242	
MUFA (Kettunen et al.)	CHD (CARDloGRAMplusC4D)	63746	
Omega 3 (UKB)	IHD (UKB)	22242	
Omega 3 (Kettunen et al.)	CHD (CARDloGRAMplusC4D)	63746	
Omega 6 (UKB)	IHD (UKB)	22242	
Omega 6 (Kettunen et al.)	CHD (CARDloGRAMplusC4D)	63746	
PUFA (UKB)	IHD (UKB)	22242	
PUFA (Kettunen et al.)	CHD (CARDloGRAMplusC4D)	63746	
SFA (UKB)	IHD (UKB)	22242	
SFA (Kettunen et al.)	CHD (CARDloGRAMplusC4D)	63746	
Total FA (UKB)	IHD (UKB)	22242	
Total FA (Kettunen et al.)	CHD (CARDIoGRAMplusC4D)	63746	
			I I I 0.75 1.0 1.5 2.0 Odds ratio per 1 mmol/L increase in exposure

Cardiovascular diseases

MR-IVW MR-WM MR-Egger MR-PRESSO

Fig 5. Two-sample MR effect estimates per 1 mmol/L higher fatty acids for different CVD outcomes in (a) UKB (angina pectoris, coronary atherosclerosis, ischemic heart disease) and (b) using Kettunen and colleagues' [20] betas for fatty acids and CARDIOGRAMplusC4D [22] for ischemic heart disease. Estimates of all examined fatty acids are shown for comparison. CVD, cardiovascular disease; DHA, docosahexaenoic acid; LA, linoleic acid; MUFA, monounsaturated fatty acid; PUFA, polyunsaturated fatty acid; SFA, saturated fatty acid; total FA, total fatty acid; UKB, UK Biobank; WM, weighted-median.

https://doi.org/10.1371/journal.pmed.1004141.g005

use of fatty acids for CVD prevention. Our results also indicate negative associations of higher DHA and omega-3 fatty acid levels with lower risk of cholelithiasis and obesity.

Extensive MR analyses supported a protective causal effect of omega-3, and its main bioactive component DHA, on the combined endpoint of cholelithiasis and cholecystitis with replication in independent databases. Several lines of evidence further support this observation. Cholelithiasis involves the presence of gallstones the vast majority of which are cholesterol stones and primarily composed of cholesterol. Long-term intake of *cis*-unsaturated and monounsaturated fats has been associated with lower risk of gallstone formation in observational settings [39], while in patients with gallstones, omega-3 fatty acid supplementation has been shown to reduce the cholesterol saturation index in bile [40]. Omega-3 also prevented gallstone formation in mice being on a lithogenic diet by increasing the levels of bile phospholipids and suppressing bile mucin formation [41]. Interestingly, analyses did not support causal associations between omega-6 fatty acid levels and cholelithiasis, suggesting that the protective effect of PUFA on the dissolution of bile disorders may occurs mainly via DHA. Overall, these results support the prioritisation of RCTs examining the role of primary and secondary prevention of cholelithiasis via dietary modification or supplementation of DHA and other omega-3 PUFAs.



Fig 6. Two-sample MR effect estimates per 1 mmol/L higher fatty acids for obesity in (a) UKB and (b) using Kettunen and colleagues' [20] betas for fatty acids and betas from GIANT [21] for BMI. Only associations with statistically significant (P < 0.05) IVW MR estimates in the discovery phase are shown. BMI, body mass index; DHA, docosahexaenoic acid; IVW, inverse-variance weighted; MR, mendelian randomisation; MUFA, monounsaturated fatty acid; total FA, total fatty acid; UKB, UK Biobank; WM, weighted-median.

https://doi.org/10.1371/journal.pmed.1004141.g006

A protective potentially causal effect of genetically predicted omega-3 fatty acids and especially DHA on obesity was also observed. Several animal and human studies have suggested an association of omega-3 fatty acids on lower adiposity through their inverse association with inflammatory cytokines and leptin [42]. However, the evidence on the beneficial effects of omega-3 supplementation in weight loss or body fat mass in RCTs has, to date, been inconsistent [43]. Our findings support the contribution of omega-3 and DHA in healthy weight status and prevention of obesity compared to other fatty acids; however, further evidence on the actual dose effects and the heterogeneity of effects across different fatty acids is needed to validate these observations.

There has been considerable debate on the higher PUFA levels potentially reducing CVD risk. Omega-3 supplements are widely used for CVD prevention despite lack of support from regulatory bodies at least for secondary prevention [44,45]. A large body of observational studies has supported an association between omega-3—mainly eicosapentaenoic acid (EPA) and DHA—and cardiometabolic diseases, such as ischemic heart disease and T2D [46–49]. However, several RCTs have failed to show an indisputable benefit of omega-3 supplementation on a range of CVD outcomes, while there is some suggestive evidence for a protective effect of



Fig 7. Two-sample multivariable MR effect estimates per 1 mmol/L higher fatty acids on CHD, using UKB betas for fatty acids and betas from CARDIOGRAMplusC4D [22] for CHD. The last column illustrated the MIP per fatty acid (MR-BMA output) to prioritise risk factors on CHD. Five models were fitted to investigate the direct effects of fatty acids. Model 1 considered the simultaneous effects of DHA, LA, MUFA, and SFA; Model 2 the effects of omega-3, omega-6, MUFA, and SFA; Model 3 the effects of PUFA, MUFA, and SFA; Model 4: DHA and LA; and Model 5: omega-3 and omega-6. MIP in the last column. CHD, coronary heart disease; DHA, docosahexaenoic acid; LA, linoleic acid; MIP, marginal inclusion probability; MR, mendelian randomisation; MR-BMA, MR-Bayesian method averaging; MUFA, monounsaturated fatty acid; PUFA, polyunsaturated fatty acid; SFA, saturated fatty acid; UKB, UK Biobank.

https://doi.org/10.1371/journal.pmed.1004141.g007

high-dose supplementation, which needs replication and further evaluation [4,50]. The present MR findings further augment the conflicting evidence base. In agreement with previous work [51], we found no evidence of protective effect between genetically predicted omega-3 fatty acids and CHD. Our results provided evidence for a possible causal detrimental effect of LA and omega-6 levels in relation to CHD and associated CVD outcomes. These associations remained significant after excluding SNPs in the FADS locus; however, horizontal pleiotropy could still explain these associations. Multivariable MR and MR-BMA analyses indicated LA as having an independent effect on CVD and as the risk factor with the strongest overall evidence for positive association with CHD. Evidence from observational studies on dietary omega-6 fatty acids supports reverse associations with incident CVD [52], while evidence from RCTs showed null associations [53]. The apparent controversy between different lines of evidence may, to a certain extent, be explained by the different definitions of fatty acids between traditional observational studies (mainly dietary fatty acids), RCTs (fatty acid supplementation), and MR (genetically predicted fatty acid levels). Additionally, MR represents the

lifetime effects of the genetic variants rather than a short period of intervention usually examined in RCTs. Nonetheless, the present evidence highlighted the need for further research on the effects of different fatty acids on CVD and does not support the otherwise wide use of fatty acid supplements for CVD prevention.

Strengths and limitations

Our study has notable strengths. First, it benefits from the largest available population data, which allowed us to assess potential sources of heterogeneity in MR findings and to replicate our results in external databases increasing the generalizability and validity of our findings. Additionally, we examined potential beneficial and detrimental effects of fatty acids across the phenome, and we examined 8 different fatty acid categories with multivariable analyses to establish their relative importance extending previous work on PUFA and cardiovascular outcomes [52]. However, our results must be interpreted in the context of a few limitations. Our work was confined to individuals of European descent, thus limiting the amenability of our findings to other ethnic groups. Also, we were unable to assess the effect of fatty acid ratios such as omega-6-to-omega-3 fatty acid that might be important for some of the examined clinical outcomes. Although our study strengthens the case for a causal relationship between fatty acids and several diseases, we cannot prove causality, as this is an exploratory hypothesis generating study that requires further validation in the form of a clinical study. For example, pleiotropy is always a potential problem in MR and occurs when a genetic instrument is associated with the outcome through pathways not including the exposure of interest. We used extensive sensitivity analyses to account for pleiotropy and implemented a Bayesian multivariable approach to select among the studied fatty acids the one more likely to be causal. This is important as the examined fatty acids have high genetic correlation. Therefore, the observed associations could reflect an association between a fatty acid other than those we have investigated, or even a combination of fatty acids. Lastly, exogenous factors may affect the genetic associations with the fatty acids, since 13,577 participants with available metabolic profiles in UKB were omega-3 supplementation and/or lipid lowering drug users. To address this, we recalculated the associations between the genetic variants used as instruments in the analysis and the 8 fatty acid measurements and observed a percentage change in the effect size of the genetic associations smaller than 1% for most variants and smaller than 10% for all, indicating that our findings are not biased by the inclusion of medication or supplementation users.

Conclusions

We have systematically investigated the phenotypic consequences of different fatty acids. We present evidence for a potentially protective effect of omega-3 fatty acid levels and DHA in particular on cholelithiasis, highlighting the need to assess them in clinical trials. At the same time, our findings do not support the supplementation of unsaturated fatty acids for CVD prevention. We found evidence for positive associations of omega-6 fatty acids and LA in particular with a range of CVD outcomes further enhancing the conflicting evidence based on fatty acids supplementation for CVD prevention and treatment.

Supporting information

S1 STROBE Checklist. STROBE Checklist. (DOCX)

S1 Table. (a) Descriptive characteristics of the UK Biobank samples used in the discovery phase of the study. UK Biobank participants were split into two datasets: the base dataset,

where the fatty acid measurements were obtained, and the target dataset, which was used to extract information on the clinical diagnoses (outcome data). (b) Descriptive characteristics of the independent cohorts used in the replication phase. (XLSX)

S2 Table. Phenotypic and genetic correlation matrices between circulating fatty acid measurements in 117,143 UK Biobank samples. Genetic correlations were estimated using LD score regression.

(XLSX)

S3 Table. (a) List of independent genetic variants used as instruments per exposure in the MR-PheWAS analysis. (b) Gene annotation of the genetic variants used as instruments in the MR-PheWAS analysis.

(XLSX)

S4 Table. F-statistic of the associations between genetic variants and fatty acids. (XLSX)

S5 Table. Number of health outcomes and cases per disease group included in the analysis. Health outcomes with more than 200 cases were considered. (XLSX)

S6 Table. Phenome-wide association study results of docosahexaenoic acid (DHA) genetic risk score.

(XLSX)

S7 Table. Phenome-wide association study results of linoleic acid (LA) genetic risk score. (XLSX)

S8 Table. Phenome-wide association study results of monounsaturated fatty acids (MUFAs) genetic risk score. (XLSX)

S9 Table. Phenome-wide association study results of omega-3 fatty acids genetic risk score. (XLSX)

S10 Table. Phenome-wide association study results of omega-6 fatty acids genetic risk score.

(XLSX)

S11 Table. Phenome-wide association study results of polyunsaturated fatty acids (PUFAs) genetic risk score.

(XLSX)

S12 Table. Phenome-wide association study results of saturated fatty acids (SFAs) genetic risk score.

(XLSX)

S13 Table. Phenome-wide association study results of total fatty acids genetic risk score. (XLSX)

S14 Table. Mendelian randomisation results for the effect of fatty acids on health outcomes. 114 associations were taken forward for MR analysis. 60 associations were statistically significant (P < 0.05) in MR-IVW, MR-weighted median and MR-PRESSO methods, with consistent MR-Egger estimate in the same direction as MR-IVW and nonsignificant MR-Egger intercept. (XLSX)

S15 Table. Two-sample mendelian randomisation in the replication phase using Kettunen and colleagues' betas for the fatty acids and FinnGen, CARDIoGRAMplusC4D, GIANT, MAGIC, IGAP, and DIAGRAM summary statistics for health outcomes with available summary statistics.

(XLSX)

S16 Table. Two-sample mendelian randomisation results for CHD, cholelithiasis, and BMI, after excluding genetic variants from the FADS locus. (XLSX)

S17 Table. Two-sample mendelian randomisation in the discovery phase using UKB for the fatty acids and CARDIoGRAMplusC4D summary statistics for coronary heart disease. (XLSX)

S18 Table. Multivariable mendelian randomisation and mendelian randomisation–Bayesian method averaging (MR-BMA) results for the direct effect of fatty acids on coronary heart disease using summary statistics from CARDIoGRAMplusC4D. Both methods were conducted on three separate models. Model 1: DHA, LA, MUFA, SFA; Model 2: omega-3, omega-6, MUFA, SFA; Model 3: PUFA, MUFA, SFA; Model 4: DHA, LA; Model 5: omega-3, omega-6.



S1 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against docosahexaenoic acid (DHA) genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S2 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against linoleic acid (LA) genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S3 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against monounsaturated fatty acids (MUFA) genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S4 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against omega-3 fatty acids genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S5 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against omega-6 fatty acids genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S6 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against polyunsaturated fatty acids (PUFA) genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S7 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against saturated fatty acids (SFA) genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

S8 Fig. PheWAS Manhattan plot. 845 clinical diagnoses were regressed against total fatty acids genetic risk score (GRS). Age, sex, and the first 10 genetic principal components were used as covariates in the logistic regressions. (DOCX)

Author Contributions

Conceptualization: Abbas Dehghan, Ioanna Tzoulaki.

Data curation: Loukas Zagkos, Marie-Joe Dib.

Formal analysis: Loukas Zagkos, Marie-Joe Dib.

Funding acquisition: Ioanna Tzoulaki.

Investigation: Marie-Joe Dib, Abbas Dehghan, Ioanna Tzoulaki.

Methodology: Loukas Zagkos, Marie-Joe Dib, Dipender Gill, Fotios Koskeridis, Fotios Drenos, Verena Zuber, Kostas Tsilidis, Abbas Dehghan, Ioanna Tzoulaki.

Project administration: Loukas Zagkos, Paul Elliott, Abbas Dehghan, Ioanna Tzoulaki.

Resources: Paul Elliott.

Software: Rui Pinto, Verena Zuber.

Supervision: Paul Elliott, Abbas Dehghan, Ioanna Tzoulaki.

Validation: Loukas Zagkos.

Visualization: Loukas Zagkos, Rui Pinto.

Writing - original draft: Loukas Zagkos, Marie-Joe Dib, Ioanna Tzoulaki.

Writing – review & editing: Loukas Zagkos, Marie-Joe Dib, Rui Pinto, Dipender Gill, Fotios Koskeridis, Fotios Drenos, Georgios Markozannes, Paul Elliott, Verena Zuber, Kostas Tsilidis, Abbas Dehghan, Ioanna Tzoulaki.

References

- 1. Nagy K, Tiuca I. Importance of Fatty Acids in Physiopathology of Human Body. 2017.
- Fritsche KL. The science of fatty acids and inflammation. Adv Nutr. 2015; 6(3):293S–301S. Epub 2015/ 05/17. https://doi.org/10.3945/an.114.006940 PMID: 25979502; PubMed Central PMCID: PMC4424767.
- Calder PC. Fatty acids and inflammation: the cutting edge between food and pharma. Eur J Pharmacol. 2011; 668(Suppl 1):S50–8. Epub 2011/08/06. https://doi.org/10.1016/j.ejphar.2011.05.085 PMID: 21816146.
- Rizos EC, Markozannes G, Tsapas A, Mantzoros CS, Ntzani EE. Omega-3 supplementation and cardiovascular disease: formulation-based systematic review and meta-analysis with trial sequential analysis. Heart. 2021; 107(2):150–8. Epub 2020/08/21. https://doi.org/10.1136/heartjnl-2020-316780 PMID: 32820013.
- Markozannes G, Ntzani EE, Tsapas A, Mantzoros CS, Tsiara S, Xanthos T, et al. Dose-related metaanalysis for Omega-3 fatty acids supplementation on major adverse cardiovascular events. Clin Nutr. 2022; 41(4):923–930. https://doi.org/10.1016/j.clnu.2022.02.022 WOS:000781075400012 PMID: 35290840

- von Schacky C. Meta-analysing randomised controlled trials with omega-3 fatty acids in cardiovascular disease. Evid Based Med. 2013; 18(4):e33. Epub 2012/11/20. https://doi.org/10.1136/eb-2012-100960 PMID: 23161504.
- Alexander DD, Miller PE, Van Elswyk ME, Kuratko CN, Bylsma LC. A Meta-Analysis of Randomized Controlled Trials and Prospective Cohort Studies of Eicosapentaenoic and Docosahexaenoic Long-Chain Omega-3 Fatty Acids and Coronary Heart Disease Risk. Mayo Clin Proc. 2017; 92(1):15–29. Epub 2017/01/08. https://doi.org/10.1016/j.mayocp.2016.10.018 PMID: 28062061.
- O'Mahoney LL, Matu J, Price OJ, Birch KM, Ajjan RA, Farrar D, et al. Omega-3 polyunsaturated fatty acids favourably modulate cardiometabolic biomarkers in type 2 diabetes: a meta-analysis and metaregression of randomized controlled trials. Cardiovasc Diabetol. 2018; 17(1):98. Epub 2018/07/10. https://doi.org/10.1186/s12933-018-0740-x PMID: 29981570; PubMed Central PMCID: PMC6035402.
- Black LI CT, Barnes PM, Stussman BJ, Nahin RL. Use of complementary health approaches among children aged 4–17 years in the United States: National Health Interview Survey. Natl Health Stat Rep. 2015; 10(78):1–19. PubMed Central PMCID: PMC4562218.
- Clarke TC. The use of complementary health approaches among US cancer survivors. Cancer Res. 2015; 75. https://doi.org/10.1158/1538-7445.Am2015-3723 WOS:000371597102260.
- Lemaitre RN, Tanaka T, Tang W, Manichaikul A, Foy M, Kabagambe EK, et al. Genetic loci associated with plasma phospholipid n-3 fatty acids: a meta-analysis of genome-wide association studies from the CHARGE Consortium. PLoS Genet. 2011; 7(7):e1002193. Epub 20110728. https://doi.org/10.1371/ journal.pgen.1002193 PMID: 21829377; PubMed Central PMCID: PMC3145614.
- Guan WH, Steffen BT, Lemaitre RN, Wu JHY, Tanaka T, Manichaikul A, et al. Genome-Wide Association Study of Plasma N6 Polyunsaturated Fatty Acids Within the Cohorts for Heart and Aging Research in Genomic Epidemiology Consortium. Circ-Cardiovasc Gene. 2014; 7 (3):321–331. <u>https://doi.org/10. 1161/CIRCGENETICS.113.000208 WOS:000338207300013 PMID: 24823311</u>
- 13. Wu JH, Lemaitre RN, Manichaikul A, Guan W, Tanaka T, Foy M, et al. Genome-wide association study identifies novel loci associated with concentrations of four plasma phospholipid fatty acids in the de novo lipogenesis pathway: results from the Cohorts for Heart and Aging Research in Genomic Epidemiology (CHARGE) consortium. Circ Cardiovasc Genet. 2013; 6(2):171–83. Epub 20130129. https://doi.org/10.1161/CIRCGENETICS.112.964619 PMID: 23362303; PubMed Central PMCID: PMC3891054.
- Shin SY, Fauman EB, Petersen AK, Krumsiek J, Santos R, Huang J, et al. An atlas of genetic influences on human blood metabolites. Nat Genet. 2014; 46(6):543–50. Epub 20140511. https://doi.org/10.1038/ ng.2982 PMID: 24816252; PubMed Central PMCID: PMC4064254.
- Tintle NL, Pottala JV, Lacey S, Ramachandran V, Westra J, Rogers A, et al. A genome-wide association study of saturated, mono- and polyunsaturated red blood cell fatty acids in the Framingham Heart Offspring Study. Prostaglandins Leukot Essent Fatty Acids. 2015; 94:65–72. Epub 20141202. https://doi. org/10.1016/j.plefa.2014.11.007 PMID: 25500335; PubMed Central PMCID: PMC4339483.
- Sudlow C, Gallacher J, Allen N, Beral V, Burton P, Danesh J, et al. UK biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age. PLoS Med. 2015; 12(3):e1001779. Epub 2015/04/01. https://doi.org/10.1371/journal.pmed.1001779 PMID: 25826379; PubMed Central PMCID: PMC4380465.
- Bycroft C, Freeman C, Petkova D, Band G, Elliott LT, Sharp K, et al. The UK Biobank resource with deep phenotyping and genomic data. Nature. 2018; 562(7726):203–9. Epub 2018/10/12. https://doi. org/10.1038/s41586-018-0579-z PMID: 30305743; PubMed Central PMCID: PMC6786975.
- Elsworth B, Lyon M., Alexander T., Liu Y., Matthews P., Hallett J. et al. The MRC IEU OpenGWAS data infrastructure. BioRxiv [preprint]. 2020. https://doi.org/10.1101/2020.08.10.244293
- 19. Purcell SNB, Todd-Brown K, Thomas L, Ferreira MAR, Bender D, Maller J, et al. PLINK: a toolset for whole-genome association and population-based linkage analysis. Am J Hum Genet. 2007.
- Kettunen J, Demirkan A, Wurtz P, Draisma HH, Haller T, Rawal R, et al. Genome-wide study for circulating metabolites identifies 62 loci and reveals novel systemic effects of LPA. Nat Commun. 2016; 7:11122. Epub 2016/03/24. https://doi.org/10.1038/ncomms11122 PMID: 27005778; PubMed Central PMCID: PMC4814583.
- Yengo L, Sidorenko J, Kemper KE, Zheng Z, Wood AR, Weedon MN, et al. Meta-analysis of genomewide association studies for height and body mass index in approximately 700000 individuals of European ancestry. Hum Mol Genet. 2018; 27(20):3641–9. Epub 2018/08/21. https://doi.org/10.1093/hmg/ ddy271 PMID: 30124842; PubMed Central PMCID: PMC6488973.
- Nikpay M, Goel A, Won HH, Hall LM, Willenborg C, Kanoni S, et al. A comprehensive 1,000 Genomesbased genome-wide association meta-analysis of coronary artery disease. Nat Genet. 2015; 47 (10):1121–30. Epub 2015/09/08. <u>https://doi.org/10.1038/ng.3396</u> PMID: <u>26343387</u>; PubMed Central PMCID: PMC4589895.

- Lambert JC, Ibrahim-Verbaas CA, Harold D, Naj AC, Sims R, Bellenguez C, et al. Meta-analysis of 74,046 individuals identifies 11 new susceptibility loci for Alzheimer's disease. Nat Genet. 2013; 45 (12):1452–8. Epub 2013/10/29. https://doi.org/10.1038/ng.2802 PMID: 24162737; PubMed Central PMCID: PMC3896259.
- Soranzo N, Sanna S, Wheeler E, Gieger C, Radke D, Dupuis J, et al. Common Variants at 10 Genomic Loci Influence Hemoglobin A(1C) Levels via Glycemic and Nonglycemic Pathways. Diabetes. 2010; 59 (12):3229–3239. https://doi.org/10.2337/db10-0502 WOS:000285988200032 PMID: 20858683
- 25. Mahajan A, Go MJ, Zhang WH, Below JE, Gaulton KJ, Ferreira T, et al. Genome-wide trans-ancestry meta-analysis provides insight into the genetic architecture of type 2 diabetes susceptibility. Nat Genet. 2014; 46(3):234–+. https://doi.org/10.1038/ng.2897 WOS:000332036700007. PMID: 24509480
- Benjamini Y, Hochberg Y. Controlling the False Discovery Rate—a Practical and Powerful Approach to Multiple Testing. J R Stat Soc B 1995; 57(1):289–300. <u>https://doi.org/10.1111/j.2517-6161.1995</u>. tb02031.x WOS:A1995QE45300017.
- Burgess S, Butterworth A, Thompson SG. Mendelian Randomization Analysis With Multiple Genetic Variants Using Summarized Data. Genet Epidemiol. 2013; 37(7):658–665. <u>https://doi.org/10.1002/gepi.21758</u> WOS:000325980600002. PMID: 24114802
- Bowden J, Smith GD, Burgess S. Mendelian randomization with invalid instruments: effect estimation and bias detection through Egger regression. Int J Epidemiol. 2015; 44 (2):512–525. https://doi.org/10. 1093/ije/dyv080 WOS:000357106100015. PMID: 26050253
- Bowden J, Smith GD, Haycock PC, Burgess S. Consistent Estimation in Mendelian Randomization with Some Invalid Instruments Using a Weighted Median Estimator. Genet Epidemiol. 2016; 40 (4):304– 314. https://doi.org/10.1002/gepi.21965 WOS:000374542600005. PMID: 27061298
- Verbanck M, Chen CY, Neale B, Do R. Detection of widespread horizontal pleiotropy in causal relationships inferred from Mendelian randomization between complex traits and diseases. Nat Genet. 2018; 50(5):693–8. Epub 2018/04/25. https://doi.org/10.1038/s41588-018-0099-7 PMID: 29686387; PubMed Central PMCID: PMC6083837.
- Burgess S, Dudbridge F, Thompson SG. Multivariable Mendelian Randomization: The Use of Pleiotropic Genetic Variants to Estimate Causal Effects. Am J Epidemiol. 2015; 181(4):290–291. <u>https://doi.org/10.1093/aje/kwv017</u> WOS:000351501400011. PMID: 25660081
- Zuber V, Colijn JM, Klaver C, Burgess S. Selecting likely causal risk factors from high-throughput experiments using multivariable Mendelian randomization. Nat Commun. 2020; 11(1):29. Epub 2020/01/09. https://doi.org/10.1038/s41467-019-13870-3 PMID: <u>31911605</u>; PubMed Central PMCID: PMC6946691.
- Sanderson E, Smith GD, Windmeijer F, Bowden J. An examination of multivariable Mendelian randomization in the single-sample and two-sample summary data settings. Int J Epidemiol. 2019; 48 (3):713– 727. https://doi.org/10.1093/ije/dyy262 WOS:000486642400012. PMID: 30535378
- Team RC. R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing; 2020.
- Carroll RJ, Bastarache L, Denny JC. R PheWAS: data analysis and plotting tools for phenome-wide association studies in the R environment. Bioinformatics. 2014; 30(16):2375–6. Epub 2014/04/16. https://doi.org/10.1093/bioinformatics/btu197 PMID: 24733291; PubMed Central PMCID: PMC4133579.
- Hemani G, Zheng J, Elsworth B, Wade KH, Haberland V, Baird D, et al. The MR-Base platform supports systematic causal inference across the human phenome. Elife. 2018; 7. Epub 2018/05/31. https://doi.org/10.7554/eLife.34408 PMID: 29846171; PubMed Central PMCID: PMC5976434.
- Gordon M, Lumley, T., Gordon, M. M. R Package 'forestplot'. Advanced Forest Plot Using 'grid' Graphics. 2016.
- Bulik-Sullivan BK, Loh PR, Finucane HK, Ripke S, Yang J, Schizophrenia Working Group of the Psychiatric Genomics C, et al. LD Score regression distinguishes confounding from polygenicity in genomewide association studies. Nat Genet. 2015; 47(3):291–5. Epub 2015/02/03. https://doi.org/10.1038/ng. 3211 PMID: 25642630; PubMed Central PMCID: PMC4495769.
- Tsai CJ, Leitzmann MF, Willett WC, Giovannucci EL. The effect of long-term intake of cis unsaturated fats on the risk for gallstone disease in men—A prospective cohort study. Ann Intern Med. 2004; 141 (7):514–522. https://doi.org/10.7326/0003-4819-141-7-200410050-00007 WOS:000224467800003. PMID: 15466768
- 40. Berr F, Holl J, Jungst D, Fischer S, Richter WO, Seifferth B, et al. Dietary N-3 Polyunsaturated Fatty-Acids Decrease Biliary Cholesterol Saturation in Gallstone Disease. Hepatology. 1992; 16(4):960–967. https://doi.org/10.1002/hep.1840160418 WOS:A1992JR88300017. PMID: 1398503
- 41. Kim JK, Cho SM, Kang SH, Kim E, Yi H, Yun ES, et al. N-3 polyunsaturated fatty acid attenuates cholesterol gallstones by suppressing mucin production with a high cholesterol diet in mice. J Gastroenterol

Hepatol. 2012; 27 (11):1745–1751. https://doi.org/10.1111/j.1440-1746.2012.07227.x WOS:000310554000017. PMID: 22849613

- 42. Gray B, Steyn F, Davies PSW, Vitetta L. Omega-3 fatty acids: a review of the effects on adiponectin and leptin and potential implications for obesity management. Eur J Clin Nutr. 2013; 67 (12):1234– 1242. https://doi.org/10.1038/ejcn.2013.197 WOS:000327925600002. PMID: 24129365
- 43. Delpino FM, Figueiredo LM, da Silva BGC. Effects of omega-3 supplementation on body weight and body fat mass: A systematic review. Clinical Nutrition Espen. 2021; 44:122–129. <u>https://doi.org/10. 1016/j.clnesp.2021.04.023</u> WOS:000678558000013. PMID: 34330455
- 44. Agency EM. EMA confirms omega-3 fatty acid medicines are not effective in preventing further heart problems after a heart attack. 2019 [cited 09.06.2022]. Available from: https://www.ema.europa.eu/en/ medicines/human/referrals/omega-3-acid-ethyl-esters-containing-medicinal-products-oral-usesecondary-prevention-after.
- 45. Skulas-Ray AC, Wilson PWF, Harris WS, Brinton EA, Kris-Etherton PM, Richter CK, et al. Omega-3 Fatty Acids for the Management of Hypertriglyceridemia: A Science Advisory From the American Heart Association. Circulation. 2019; 140(12):E673–E691. https://doi.org/10.1161/CIR.00000000000000009 WOS:000486196500006. PMID: 31422671
- 46. Brown TJ, Brainard J, Song F, Wang X, Abdelhamid A, Hooper L, et al. Omega-3, omega-6, and total dietary polyunsaturated fat for prevention and treatment of type 2 diabetes mellitus: systematic review and meta-analysis of randomised controlled trials. Bmj-Brit Med J. 2019: 366. doi: ARTN I4697. WOS:000482882000003. https://doi.org/10.1136/bmj.I4697 PMID: 31434641
- Naini AE, Keyvandarian N, Mortazavi M, Taheri S, Hosseini SM. Effect of Omega-3 fatty acids on blood pressure and serum lipids in continuous ambulatory peritoneal dialysis patients. J Res Pharm Pract. 2015; 4(3):135–141. <u>https://doi.org/10.4103/2279-042X.162356</u> WOS:000218634800005. PMID: 26312252
- Rajaei E, Mowla K, Ghorbani A, Bahadoram S, Bahadoram M, Dargahi-Malamir M. The Effect of Omega-3 Fatty Acids in Patients With Active Rheumatoid Arthritis Receiving DMARDs Therapy: Double-Blind Randomized Controlled Trial. Global J Health Sci. 2015; 8(7):18–25. Epub 2015/01/01. https://doi.org/10.5539/gjhs.v8n7p18 PMID: 26925896; PubMed Central PMCID: PMC4965662.
- Sekikawa A, Cui CD, Sugiyama D, Fabio A, Harris WS, Zhang X. Effect of High-Dose Marine Omega-3 Fatty Acids on Atherosclerosis: A Systematic Review and Meta-Analysis of Randomized Clinical Trials. Nutrients 2019; 11(11). doi: ARTN 2599 <u>https://doi.org/10.3390/nu11112599</u> WOS:000502274600050. PMID: 31671524
- Rizos EC, Ntzani EE, Bika E, Kostapanos MS, Elisaf MS. Association between omega-3 fatty acid supplementation and risk of major cardiovascular disease events: a systematic review and meta-analysis. JAMA. 2012; 308(10):1024–33. Epub 2012/09/13. https://doi.org/10.1001/2012.jama.11374 PMID: 22968891.
- 51. Borges MC, Haycock PC, Zheng J, Hemani G, Holmes MV, Smith GD, et al. Role of circulating polyunsaturated fatty acids on cardiovascular diseases risk: analysis using Mendelian randomization and fatty acid genetic association data from over 114,000 UK Biobank participants. BMC Med. 2022; 20(1). doi: ARTN 210 https://doi.org/10.1186/s12916-022-02399-w WOS:000809666500001. PMID: 35692035
- 52. Marklund M, Wu JHY, Imamura F, Del Gobbo LC, Fretts A, de Goede J, et al. Biomarkers of Dietary Omega-6 Fatty Acids and Incident Cardiovascular Disease and Mortality: An Individual-Level Pooled Analysis of 30 Cohort Studies. Circulation. 2019; 139(21):2422–2436. https://doi.org/10.1161/ Circulationaha.118.038908 WOS:000469018300011. PMID: 30971107
- Hooper L, Al-Khudairy L, Abdelhamid AS, Rees K, Brainard JS, Brown TJ, et al. Omega-6 fats for the primary and secondary prevention of cardiovascular disease. Cochrane Db Syst Rev. 2018; (11). doi: ARTN CD01109410.1002/14651858.CD011094.pub4. WOS:000455159600006.